Author's response to reviews

Title: Social representations of male circumcision as prophylaxis against HIV/AIDS in Zimbabwe

Authors:

Antony Chikutsa (antony.chikutsa@gmail.com)
Pranitha Maharaj (maharajp7@ukzn.ac.za)

Version: 4 Date: 22 June 2015

Author's response to reviews: see over
The BMC Public Health Editorial Team

We have revised our manuscript in line with the second reviewer’s comments as follows:

Reviewer # 2: Rebecca Fielding-Miller

1. Define VMMC (lines 27-28).

Response
This was done in the abstract and also on the referred text.

2. Line 160: correct “conveniently sampled”

Response
This was done and the sentence now reads as follows:

Participants in the FGDs were randomly selected from households that were sampled in high density suburbs of Harare urban except for participants in the mixed sex FGD who were selected using convenience sampling from a local shopping area.

3. Make uniform the reporting FGD participant characteristics.

Response
This was done as recommended.